ഹ

ADVANCING EARTH AND SPACE SCIENCE

Geochemistry, Geophysics, Geosystems[°]

RESEARCH ARTICLE

10.1029/2021GC009674

Key Points:

- Joint inversion that includes USArray data allows enhanced resolution of regional upper mantle velocity models
- NA13 reveals small scale velocity variations beneath the Central and Eastern United States
- Upper mantle velocity variation is on the order of ~100 km

Correspondence to: H. Bedle, hbedle@ou.edu

Citation:

Bedle, H., Lou, X., & van der Lee, S. (2021). Continental tectonics inferred from high-resolution imaging of the mantle beneath the United States, through the combination of USArray data types. *Geochemistry, Geophysics, Geosystems, 22*, e2021GC009674. https://doi.org/10.1029/2021GC009674

Received 25 JAN 2021 Accepted 26 JUL 2021

© 2021 The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

t includes s enhanced nal upper mantle I scale velocity to Control and Control and Control and the Combination of USArray Data Types Heather Bedle¹, , Xiaoting Lou^{2,3}, and Suzan van der Lee² School of Geosciences, University of Oklahoma, Norman, OK, USA, ²Earth and Planetary Sciences, Northwestern University, Evanston, IL, USA, ³Now at Chevron Energy Technology Company, Houston, TX, USA

Abstract A comprehensive North American upper mantle seismic-tomographic model, NA13, was inferred from a combination of several decades of seismic data from early North American seismic networks and USArray data. This data-driven modeling inverted a combination of regional waveform fits, teleseismic *S* delay times, and point constraints on Moho depths. The joint inversion model combines the contrasting and overlapping resolving power of the different data sets, and demonstrates enhanced resolution over regional models created with a single geophysical data set. Resolution in the upper mantle is achieved on the scale of ~100 km, although travel time delay studies suggest that anomalies in parts of the western United States are smaller. In NA13, velocities beneath the Yellowstone plume are low enough to require the presence of partial melt. NA13 also models a variety of smaller scale velocity variations beneath the Central and Eastern United States that reveal remnant velocity anomalies related to lithospheric variations in the upper mantle between Proterozoic units.

Continental Tectonics Inferred From High-Resolution

Imaging of the Mantle Beneath the United States,

Plain Language Summary Seismic tomography has been used for decades to image the structure of the interior of the Earth. NA13 is an updated tomographic model that expands on previous models by combining multiple types of data from seismic recordings of earthquakes to allow the imaging of smaller scale features in the upper mantle beneath the United States. These methods allow NA13 to be created on a regional, continental scale, while still being able to image smaller-scale features that are typically absent in regional models. Of interest, NA13 suggests that partial melt must exist beneath Yellowstone, and reveals anomalies beneath the Central and Eastern United States related to Mesozoic and Cenozoic tectonics.

1. Introduction

1.1. Tectonic Setting

The United States portion of the long-lived part of the North American continent consists of Precambrian cratons, including Archean cratons such as the Wyoming craton, and the Proterozoic Interior Platform (Bleeker, 2003; Hoffman, 1988). Since cratonic accretion, the Cenozoic-Mesozoic Rocky Mountain Cordillera and Paleozoic Appalachian Mountains formed on the west and east sides of the craton (Figure 1), respectively. Extending further out from the relatively stable core of the continent and to the east of the Appalachian range, the Atlantic Coastal Plain province in the east is a Paleozoic passive plate margin (Bally et al., 1989). The western edge of North America has a more complex tectonic history. Its seismically and tectonically active continental margin, Mesozoic-Cenozoic orogenies, and arc volcanism are controlled by the interaction between oceanic Pacific, Kula and Farallon plates and the North American plate (Atwater, 1989). Currently, the Juan de Fuca plate, a remnant of the Farallon plate, is subducting beneath North America at the Cascadia subduction zone (Atwater, 1970), causing arc volcanism, earthquakes, and episodic tremor and slip (Rogers & Dragert, 2003). The further inland location of the Laramide orogeny and eastward migration of magmatism during the Mesozoic can be explained by the contemporaneous flattening of the Farallon slab, which was likely caused by the increase in subduction rate and slab buoyancy (Engebretson et al., 1984, 1985; Molnar & Atwater, 1978). The Tertiary extensional system of the Basin and Range is likely due to the steepening of the Farallon Plate (Coney & Reynolds, 1977; Davis, 1980). Compared to tectonically active western North America, there is minimal tectonic activity and topographic variation in the central and eastern United States. However, variations in crustal and mantle structure do exist and have been the





Figure 1. Geologic provinces and topography map of the US. Abbreviated names: CP, Colorado Plateau; CoP, Columbia Plateau; SRP, Snake River Plain; YC, Yellowstone Caldera; WB, Wyoming Basin; SU, Superior Upland; OxP, Ozark Plateaus; OuM, Ouachita Mountains; ILP, Interior Low Plateaus. The solid purple line marks the boundary between the western and eastern US.

focus of many local tomographic studies as USArray moved eastward (e.g., Bollmann et al., 2019; Levin et al., 2018; Long et al., 2019), and improved understanding of these anomalies may provide additional insight into current central and eastern earthquakes. Most large-scale tectonic episodes leave remnant signatures in the lithosphere and asthenosphere by altering the thermal and mineralogical structure. Improved tomographic modeling that employs multiple types of data sets, such as NA13, allow for the detection and resolution of these anomalies to aid in better constraining paleotectonics.

1.2. Tomographic Models

Seismological observations, particularly using the method of seismic tomography, assist us in understanding the tectonic complexity of the North American continent. Three-dimensional North American mantle velocity structures have been inverted from pre-EarthScope body waves (e.g., Grand, 1994, 2002; Ren et al., 2007; Van der Hilst et al., 1997; Zhao, 2004), surface wave data (e.g., Boschi & Ekstrom, 2002; Laske & Masters, 1998; Li & Romanowicz, 1996; Marone et al., 2007; Nettles & Dziewonski, 2008), a combination of body and surface wave data (e.g., Panning et al., 2010; Ritsema et al., 2004, 2011; Van der Lee & Frederiksen, 2005; Van der Lee & Nolet, 1997a, 1997b), or with additional geodynamic constraints such as gravity (e.g., Godey et al., 2004; Houser et al., 2008; Simmons et al., 2010) in global and continental scales. These models typically capture large-scale features, such as the strong upper mantle velocity contrast between the tectonically active western US and the relatively stable central and eastern United States, but vary considerably in how detailed variations within each region are imaged. Models that can resolve intermediate-scale features are provided by regional and local body wave models (e.g., Dueker & Yuan, 2004; Frederiksen et al., 2007; Gao et al., 2004; Humphreys & Dueker, 1994; Levander et al., 2005; Villemaire et al., 2012; Yuan & Dueker, 2005a, 2005b) and surface wave models (Darbyshire et al., 2007; Foster et al., 2020; Li, 2003; Schutt et al., 2008; Tanimoto & Sheldrake, 2002; West et al., 2004).

The unprecedented seismic data abundance since the onset of EarthScope's USArray has improved the resolving power for both the body wave models (Burdick et al., 2008, 2012; James et al., 2011; Roth et al., 2008; Schmandt & Lin, 2014; Sigloch, 2011; Sigloch et al., 2008; Simmons et al., 2012; Tian & Zhao, 2012; Tian et al., 2009; Xue & Allen, 2007, 2010) and surface wave models (Bedle & van der Lee, 2009; Levin et al., 2018; Lin & Ritzwoller, 2011; Lin et al., 2008; Liu et al., 2011; Long et al., 2019; Pollitz, 2008; Pollitz & Snoke, 2010; Schmandt & Lin, 2014; Wagner et al., 2010; Yang & Gao, 2018; Yang et al., 2008; among many others).

Further improvement in resolving North American mantle structure results from better accounting for crustal structure and the simultaneous inversion of multiple data types (Liu et al., 2011; Obrebski et al., 2011; Tian et al., 2011). The model presented here, NA13, combines body wave travel times, constraints from regional *S* and Rayleigh waveform trains, and independent Moho depth constraints to jointly invert for a three-dimensional *S*-velocity (V_s) model for the North American mantle. The joint inversion leads to better velocity structure resolution than can be achieved by any of the individual data sets, as previously demonstrated by Schmid et al. (2008) and Chang et al. (2010). In addition, model NA13 demonstrates improved lateral resolution of upper mantle features on the order of $1^\circ \times 1^\circ$, compared to other joint-inversion models which demonstrate limits of $5^\circ \times 5^\circ$ (e.g., Golos et al., 2018; Schmandt & Lin, 2014).

2. Data and Methods

2.1. Model Parameterization

The seismic velocity of the three-dimensional model is parameterized by 25 spherical shells of grid points at 25 depths (0, 2, 20, 35, 60, 90, 120, 155, 190, 230, 270, 320, 370, 420, 480, 540, 600, 670, 740, 820, 900, 1,000, 1,100, 1,300, and 1,500 km). An additional spherical shell of grid points represents the laterally varying Moho depth. The vertical spacing of the spherical shells within the mantle increases gradually with depth to accommodate decreasing resolution. A triangular tessellation method discretizes each spherical shell (Baumgardner & Frederickson, 1985; Van der Lee & Nolet, 1997b; Wang & Dahlen, 1995). The seismic velocity is defined at any point within the mantle or crust by linear interpolation of values at the six surrounding grid points, which are nodes of triangles in each of two spherical shells, right above and below the point. Each spherical shell consists of 176 triangles and 66,266 vertices, yielding 1,722,916 grid points. The triangle-side length ranges from 40 to 47 km with an average of 45 km on the surface, which is less than the TA station spacing of 75 km. The triangular grid is centered 45°N and 95°W and extends 55° in all directions from the center as shown in Figure 2a. Moho depth is defined using linear interpolation of values at the three nodes of its bounding triangle in the Moho shell. Figure 2 additionally displays the input parameters and data for NA13, with station coverage and events displayed in Figures 2b-2d. Coverage extends past the North American continent to accommodate the extent of the body and surface wave paths. In this study, 386,762 grid points are hit by teleseismic S waves, while 876,864 grid points are hit by the regional S and Rayleigh waveform fits. In total, the data samples 960,947 grid points when combining the body and surface waves.

2.2. Joint Inversion

Teleseismic *S* wave arrival times and regional *S*, multiple *S*, and Rayleigh waveform fits are jointly inverted for *S* velocity perturbations of a reference model. This joint inversion combines the resolution and sensitivities of body and surface waves to create a more accurate and detailed velocity anomaly model. Regional *S* and multi-mode Rayleigh waveform fits were measured by the method of partitioned waveform inversion (PWI; Nolet, 1990; Van der Lee & Nolet, 1997a, 1997b). Independent Moho depth measurements such as from receiver functions, active source surveys and gravity measurements are also incorporated, along with appropriate relative uncertainties, to improve resolution of the velocity structure. The joint inversion method is similar to previous work by Feng et al. (2007), Schmid et al. (2008), and Chang et al. (2010) and described in detail for this inversion in Lou (2013). The joint inversion of differing data sets provides enhanced resolution of the velocity structure because each data set can improve the resolving power of other datasets (e.g., surface waves best constrain the uppermost mantle, and anomalies not accounted for in the upper mantle can be positioned in the lower mantle by the teleseismic waves). In addition, the joint inversion of these various datasets allows for the creation of a velocity model that has superior placement and detection of smaller scale velocity anomalies where data coverage allows.

A typical seismic tomography problem is to solve a large, sparse, and mixed determined system of linear equations in the form of $G\mathbf{m} = \mathbf{d}$ where \mathbf{m} is the unknown model vector, \mathbf{d} is the known data vector, and G is the modeled sensitivity kernel matrix. The model vector, \mathbf{m} , includes the three-dimensional *S* velocity





Figure 2. Map of (a) triangular grid (red points) of a spherical shell, which represents the model space of NA13. Note that the actual triangle-side length of the grid for NA13 is 45 km, and is too small to be properly visualized at this global scale. So, a grid with the same area coverage, but twice the horizontal spacing is plotted for illustration. (b) Seismic stations for the body wave travel time data set, (c) teleseismic earthquakes and stations for the body wave travel time data set, (d) map of seismic stations (red triangles) and regional earthquakes (green circles) with regional waveform constraints (from Bedle, 2008; Bedle & van der Lee, 2009).

perturbations to a one dimensional reference model, two dimensional Moho depth perturbations to a reference value and corrections for earthquake origin times and hypocenter locations. Residuals between the observed data and data predicted by the reference model are arranged in the data vector, **d**, with the contributions from the multiple data types. Partial derivatives of these data, with respect to the model unknowns, formulate the **G** matrix. The set of equations (**Gm** = **d**) is solved for the best fitting model (**m**) using the iterative conjugate-gradient method LSQR (Paige & Saunders, 1982a, 1982b) with regularizations.

Because wave paths are irregularly distributed in the model space, the inverse problem is partly overdetermined and partly underdetermined. Therefore, regularization is added to the inversion in terms of flattening. The flattening matrix limits the spatial gradient of the velocity model, which also limits the anomaly



size to some extent. Because of this, we only use the flattening matrix in the tomographic joint inversion, rather than also adding damping equations. Details of investigation into the model regularization and resultant travel time misfit are presented in Lou (2013), but a preferred flattening multiplier of 0.5 was chosen as it balanced the trade-off between the waveform fit data's variance reduction and the model's smoothness. Investigations were also performed to determine the ideal weighting between the different data types in the inversion. Preferred weighting based on parameter testing resulted in a travel-time data weight of 1, a waveform fit weight of 2, and a weight of 8 for the point constraints on Moho depth (additional details see Lou [2013]). The RMS residuals for the combined data normalized over 500 iterations converged to about 0.25 after a few dozen of the iterations.

2.3. Input Data

As previously mentioned, different types and sets of seismic data are combined to jointly invert for seismic velocity structure beneath North America. The Crust 2.0 model (Bassin et al., 2000) was used to initially define the seven-layer one-dimensional crustal model that consisted of ice, water, soft sediments, hard sediments, upper crust, middle crust and lower crust on a $2^{\circ} \times 2^{\circ}$ global grid. The models of crustal thickness and V_{ν}/V_{ν} from Lowry and Pérez-Gussinyé (2011) and Lowry et al. (2013), both whom jointly inverted seismic receiver functions, gravity and surface heatflow measurements, were used to correct Moho depth delay times at each station. For stations not in the grid of Lowry's model, we used the Moho depth from NA04 (Van der Lee & Frederiksen, 2005). The body wave travel time data are comprised of teleseismic P and S wave absolute delay times from previous IRIS PASSCAL arrays, EarthScope's USArray, and various permanent networks as displayed in Figures 2b and 2c. The delay times were measured using AIMBAT (Lou et al., 2013). Regional S and Rayleigh wave trains have been fitted using the method of Partitioned Waveform Inversion (PWI) as first introduced by Nolet (1990). PWI methodology and algorithms were further updated to include Moho depth as a variable by Das and Nolet (1995). Van der Lee and Nolet (1997b) performed the first waveform inversion for both continental scale variations in Moho depth and upper-mantle S-velocity structure beneath North America. Model NA13 draws on a total of 9,331 waveforms from 252 regional earthquakes that were used for three-dimensional model NA07 (Bedle & van der Lee, 2009), an update of NA04 (Van der Lee & Frederiksen, 2005). An additional 1,255 waveforms from five earthquakes recorded by USArray's Transportable Array in the western United States were used to refine knowledge of the western United States (Bedle, 2008) and are included in this inversion. Lou (2013) provides additional detail on stations, events, and delay time measurements. The stations, regional, and teleseismic earthquakes are plotted in Figure 2.

3. Results

A detailed study to determine the optimal regularization parameters for the full NA13 data set was performed by Lou (2013). In that study, weighting between the data types was determined through trial inversions to minimize variance in the data. NA13 extends deeper than previous PWI North American models NA04 (Van der Lee & Frederiksen, 2005) and NA07 (Bedle & van der Lee, 2009) due to the addition of teleseismic arrival-time data in the joint inversion.

3.1. S Velocity Model NA13

Model NA13 (Figures 3 and 4) is the joint *S* velocity model inverted simultaneously from teleseismic *S* wave delay times, regional *S* and Rayleigh waveform fits, and local Moho depth constraints. The velocities in this model are plotted as a percent deviation from the IAASPI91 velocity model (Kennett & Engdahl, 1991). As a result, relatively small scale (\sim 100 km) velocity variations in the upper mantle can be more clearly imaged than when using a single data set. In the western continent, NA13 displays narrow bands of relatively higher velocity anomalies running parallel to the coast along the Cascades subduction zone between depths of 100 and 300 km. This velocity anomaly is typically interpreted as corresponding to the subducting Juan de Fuca and Gorda slabs. The Snake River Plain and the Basin and Range province are characterized by low velocity anomalies between 100 and 300 km depths. The strongest low velocity anomaly at these depths is imaged beneath the Yellowstone hotspot in the northwestern corner of Wyoming. Both the Columbia River





Figure 3. Horizontal depth slices for model NA13. Velocities are plotted throughout as a percent deviation from the IASP91 1D velocity model (Kennett & Engdahl, 1991).

Plateau, in Washington and Oregon, and the Colorado Plateau exhibit relatively high uppermost mantle velocity anomalies at 100 km depth when compared to the rest of the western United States. Surrounded by lower velocities, the high-velocity Colorado Plateau's southern segment is of a higher velocity than the northern segment at 150 and 200 km (Figure 3). A similar difference has been observed in regional models of the Colorado Plateau, including that of Liu et al. (2011), these variations lay just to the east of the proposed region of crustal delamination as proposed by Levander et al. (2011). In the Southern Rocky





Figure 4. Velocity profiles through NA13.

Mountains province, low velocities at 150 and 200 km are modeled in a north-south trend beneath the Rio Grande Rift on the eastern side of the Colorado Plateau (Figure 3).

East of the Rocky Mountains, the tectonically stable Great Plains, Central Lowlands, and adjacent smaller provinces, display higher S velocity anomalies in the upper mantle, which reflect the relatively cool and rigid cratonic lithosphere of the North American Craton (Figures 3 and 4). These high-velocity anomalies extend from the tectonically active western United States to the Coastal Plains. Most of this lithosphere is Proterozoic in age. Velocities are higher in the northern Central Lowlands and Great Plains, including the Wyoming Craton, where the lithosphere is Archean in age. Velocities appear slightly less high beneath the Yavapai and Mazatzal provinces on the southwestern and southern edges of the North American Craton. Further to the south and east, the Paleozoic Ouachita and Appalachian orogenies display lower velocity anomalies, likely reflecting their more recent tectonic stabilization.

NA13's shallow, lithospheric portion of the United States agrees at very large scales with lithospheric models created via ambient noise tomography (Bensen et al., 2009), but additionally images smaller scale features, such as high velocities related to subduction in the Pacific Northwest, and higher velocities beneath northern Idaho. The Bensen et al. (2009) model better images intra-crustal structures than NA13. NA13 models lower velocities in the uppermost mantle beneath New England, which are not observed as prominently in the Vs ambient noise models of Bensen et al. (2009) potentially because they occur relatively deep within the cool lithosphere. Another ambient noise model by Zhao et al. (2017) also reveals similar velocity structures to the large scale velocity features of NA13 such as the slower velocities beneath the Basin and Range and Rio Grande Rift, although the velocity maps created therein are smoother and are limited to larger-scale features than NA13 due to the smoothing and damping requirements in their ambient noise method.

In the transition zone and lower mantle, the magnitudes of the velocity anomalies are smaller. Relatively high velocity structures are present beneath southeastern Washington, northeastern Oregon, eastern Nevada, and western Utah in the transition zone (Figure 3). In the eastern United States, high velocity anomalies are also observed in a broad zone throughout the transition zone and down to 900 km depth. These structures are not as anomalous as those beneath the western continent. The high velocity feature dipping to the east can be observed on the left side of cross section A-A' of Figure 4. At the large scale, this feature is similar to that modeled by Sigloch (2011) using teleseismic tomography to detect V_p anomalies. Differences between the two models also exist in the uppermost mantle, such as lower velocities at the northern edge of the Coastal Plain in NA13, and significantly lower V_s beneath the Columbia River Plateau and the eastern Basin and Range province. Higher veloci-

ties extend to the south beneath Texas in NA13. In the immediate region of the transition zone and 410 km discontinuity, studies have mapped the topography of this discontinuity (Cao & Levander, 2010; Tauzin et al., 2013). At 400 km (Figure 4), NA13 reveals a relatively high velocity beneath northern Nevada around 400 km, which is in congruence with a shallower 410 discontinuity as imaged by Cao and Levander (2010), suggesting that these high velocities could be due to the same cool thermal regime responsible for shallow-ing the discontinuity in this region. Tauzin et al. (2013) also present a western US 410 km topography model, noting a similar shallow region in northern Nevada, in line with Cao and Levander (2010) and the high



velocities observed in NA13. A similar shallower topography in Tauzin et al.'s (2013) model also is imaged at the northernmost limits of their model, beneath northern Idaho, which are also in the same region as relatively high velocities in NA13. This alignment between 410 km discontinuity topography maps and the NA13 velocity model confirm that the dominant origin of seismic velocity variations is likely temperature.

3.2. Resolution Tests

To examine how well and to what scales the imaged velocity anomalies are resolved by the tomographic inversions, resolution tests are necessary. Because the applied regularization would render a biased model covariance matrix, the data's resolving power is tested for structures of general and particular interest. A typical approach to quantify resolution is the checkerboard test. A series of typical checkerboard resolution tests are conducted on hypothetical models using the same inversion parameters (including regularization and data type weighting) as for model NA13. As displayed in the left columns of Figures 5 and 6, two hypothetical checkerboard models are created by placing a $\pm 200 \text{ m/s}$ Vs perturbation into repeating rectangular boxes of sizes of 300 and 100 km. For every input model, the test outputs are plotted in three horizontal slices are at depths to display the input velocity anomalies and for testing the data's ability to recover the presence of these velocity anomalies. The other horizontal slice is for a depth without any input velocity anomaly to test the data's ability to recover the absence of a velocity anomaly.

The first checkerboard test (Figure 5) is for velocity anomaly boxes of size $300 \times 300 \times 300$ km. These velocity heterogeneities are well recovered throughout the upper mantle, and through most of the lower mantle, particularly beneath the continental US, where most of the seismic stations are located. The horizontal resolution becomes less ideal to the east of the -90° meridian due to poor station and data coverage. Velocities are smeared across the US borders with Canada and Mexico, and across the east and west coasts. This is also due to relatively abrupt termination of USArray in these regions at the time. The resolution in the upper mantle is better than in the lower mantle because the teleseismic body waves have more crossing paths in the upper mantle than in the lower mantle. Some smearing of the velocity anomalies extends into the transition zone, where no input velocities exist in the resolution model. This smearing is minor and less than 0.5% of the total *S* velocity.

The second resolution test has boxes of the size of $100 \times 100 \times 100$ km (Figure 6). At depths of 300 and 500 km, the models show strong smearing along the southwest-northeast direction. This is because the majority of the teleseismic earthquakes are from the back azimuth of northwest and southeast (Figure 2). Despite the higher degree of smearing, this checkerboard test does reveal that velocity anomalies of the 100 km size should be detectable, and resolved, particularly in the upper mantle and beneath the western half of the United States. Additional resolution tests for NA13 are presented in Lou (2013), and demonstrate that the resolution for the lower mantle beneath the western and central US is ~200 km. Due to data coverage for NA13 favoring the western and central USA, the eastern US has a resolution of 300–500 km in the upper mantle, and minimal resolution in the lower mantle of this region.

3.3. Predicted Delays Through NA13

Tomographic models can be assessed by predicting delay times of seismic data though the velocity model and comparing them to the observed delay times. Figure 7 shows the station-average observed delays for NA13. NA13 has the ability to recreate observed *S* delay time patterns within the western and central United States. In the eastern US, NA13 also performs well in the regions with good data coverage. Interestingly, despite high station and data coverage, NA13 has some difficulty in recovering the correct delay times in California, beneath northwestern Wyoming, and in the regions encircling the Colorado Plateau. While discussed in more detail in Lou and van der Lee (2014), this could be attributed to the idea that while NA13 has high resolution of ~100 km in this region, the velocity anomalies are actually of an even smaller range, causing the misfit of delay times.





Figure 5. Resolution test for model NA13. A V_s anomaly of 200 m/s is applied to rectangular boxes of 300 km in both horizontal and vertical directions.





Figure 6. Resolution test for model NA13. A S velocity anomaly of 200 m/s is applied to rectangular boxes of 100 km in both horizontal and vertical directions.





Figure 7. Seismic profiles of V_s through tectonic features of the western United States. SA stands for the Sitezia accretion, and WM for the location of the Wallowa Mountain, as discussed in the text.





Figure 8. NA13 profiles though the Rocky Mountain Front. The locations of the Aspen anomaly (AA), San Juan Mountains (SJ), and Jemez lineament (JL) are labeled.

4. Discussion

S velocity anomalies are examined in more detail for the jointly inverted model, NA13, with a focus on the tectonic relevance. Figure 4 displays two horizontal slices at 200, 400 km depths, and two cross-sectional profiles across the North American continent. Figures 8 and 9 display additional depth slices and cross-sectional profiles through NA13 to highlight mantle velocity anomalies of particular interest.

4.1. Pacific Northwest Subduction

In the northwestern US, model NA13 images a north-to-south elongated strong high velocity band along the Cascades at 200 km depth. Profile AA' through the southern tip of this anomaly shows to be an east-dipping velocity anomaly likely representing the subducting Juan de Fuca plate. This high-velocity anomaly extends to the transition zone where it appears to flatten beneath the Basin and Range. Profile BB' cuts through these anomalies at a different azimuth, and suggests a possible continuity with high-velocity anomalies at about 800 km depth, in agreement with other regional tomographic models (e.g., Schmandt & Humphreys, 2010). This higher velocity features does not tend to be as clearly imaged in global scale *S*-velocity models, including Simmons et al. (2010), Durand et al. (2017), and Lu et al. (2019). High velocity structures





Figure 9. Central and Eastern seismic velocities beneath the United States. DA = Delaware Aulacogen. Recall that horizontal resolution of NA13 lessens to the east of 90°W.

are also imaged in the lower mantle beneath provinces east of the Great Plains (Figure 3). These high velocity anomalies are identified as subducted Farallon lithosphere as originally noted by Grand (1994). The high velocities are caused by thermal differences between the lithospheric slab and the surrounding lower mantle material. Our images of the subducted Farallon plate are largely consistent with previous

tomographic studies (Burdick et al., 2008, 2012; Grand, 2002; James et al., 2011; Obrebski et al., 2010, 2011; Schmandt & Humphreys, 2010; Shearer & Buehler, 2019; Sigloch et al., 2008; Tian et al., 2009, 2011; Van der Lee & Nolet, 1997a).

Figure 7 additionally focuses in on slices and transects beneath the western portion of NA13 in the region of the subducting slabs. Profile CC' shows the high-velocity slab signature beneath Washington, Oregon, and northern California. The southern edge of the high-velocity slab contrasts sharply with low upper-mantle velocities in the slab window to its south. The 4% velocity transition occurs over less than 100 km. Profiles DD', EE', and FF' are perpendicular to CC' and show more segmentation of the high-velocity slab anomalies, including complex fragments in the transition zone. Profile EE' is located where subducting lithosphere is the youngest and where the Gorda Ridge nearly meets the trench. This profile barely shows a high-velocity slab-related anomaly in the mantle above the transition zone. Previous tomographic models have also imaged a gap in the high-velocity slab-related anomaly (Obrebski et al., 2011; Schmandt & Lin, 2014). This gap has been proposed to be due to slab break-off (Obrebski et al., 2011; Roth et al., 2008), but more recent studies of receiver functions (Tauzin et al., 2016) do image reflections in the region, suggesting the subducted slab is present, but not easily detected in previous tomographic models, possibly due its young age upon subduction. Interestingly, the region of this weaker slab corresponds to a region of faster continental crust, and less seismic tremor based on a study by Delph et al. (2018). In this region, Delph et al. (2018) suggest that their data can be explained by a lower fluid content in the slab, or due to a lower fluid permeability that is not allowing the mantle wedge to be serpentinized. The lower seismic velocities observed in NA13 are slightly north of where the Gorda Ridge nearly meets the trench, which is consistent with the oblique convergence of the Juan De Fuca and Gorda plates with the North American Plate.

In addition, NA13 displays a high-velocity anomaly beneath the Wallowa Mountains of the Columbia Plateau, similar to that imaged by Schmandt and Humphreys (2011). It is possible that this anomaly is a western promontory part the Precambrian North American lithosphere, given its connection to cratonic lithosphere shown in the right half of profile DD'. Alternatively, this anomaly could represent a phase of Farallon or Kula subduction from before 40 Ma, given its connection to a northern transition zone high-velocity anomaly, as shown in the center of profile DD', that extends upwards to a depth of ~250 km. We interpret the southern transition zone high-velocity anomaly, seen in EE' and FF" as more recently subducted fragments of the Farallon Plate. More precisely, a ~1% high-velocity anomaly resides between 300 and 500 km depth range beneath the Snake River Plain and the northern Basin and Range (Figure 3), and is imaged in the EE', FF", GG', HH', and II' cross-sections. Other *S* velocity models, such as Van der Lee and Nolet's (1997a) and Sigloch et al.'s (2008) image a similar feature, interpreted as related to Farallon subduction.

First noted by Zandt et al. (2004), a small high velocity lithospheric anomaly is observed in southern California (to the east of C' in Figure 7). This anomaly, referred to as the South Sierra Nevada anomaly is observed by Obrebski et al. (2011) in their joint inversion extending to 400 km. In NA13, this anomaly appears to extend to ~350 km depth. The lateral resolution of the Golos et al. (2018) model is limited to features larger than 400 km, and indeed this model does not show this feature.

4.2. Yellowstone Region

The Yellowstone hotspot and its related thermal anomalies are another significant component of the recent modification western United States' crust and mantle (Humphreys et al., 2000). Profiles GG', HH', and II' examine the upper mantle of the Yellowstone region. Here, strong low velocity anomalies are present in the upper mantle from the Basin and Range to the Snake River Plain. These low-velocity anomalies are strongest in the upper 200 km beneath Yellowstone as mapped in GG'. The maximum slow anomaly occurs right beneath the Yellowstone Caldera and exceeds a 10% anomaly at a depth of ~60 km. This maximum perturbation value is similar to that of Schmandt and Humphreys (2010) (-11% for V_s) and significantly larger than the -7% of Tian et al. (2011)'s. At a depth of 30 km, model NA13 displays an even larger peak low velocity of -13.8%, some of which might be attributable to velocity modeling interference from crustal structure. According to Cammarano et al. (2003), a -14% S velocity anomaly requires a temperature anomaly of 700 K if it is purely caused by heat. Such high temperature is above the solidus of peridotite at 50–100 km depth range (Goes & van der Lee, 2002; Thompson, 1992). Therefore, it is likely that partial

melt occurs beneath the Yellowstone Caldera and the easternmost Snake River Plain, which is linked to magmatism in the area. The presence of partial melt can be further noted by investigating *P*-velocity anomalies. As NA13 is a *S*-velocity model, Lou (2013)'s XL13P *P*-velocity model is queried, revealing a -4.5% V_p anomaly in this region, just slightly less of an anomaly then the -5%-7% *P*-wave velocity noted by Huang et al.'s (2014) local Yellowstone regional study, which estimated a melt fraction of ~9% in the lower crust, by following the methods of Chu et al. (2010). Calculating partial melt based solely on seismic velocities includes assumptions regarding density, temperature, and attenuation. Several studies have used seismic velocity data to attempt to quantify partial melt deeper in the earth (Goes & van der Lee, 2002; Schutt & Humphreys, 2004), as compared to Huang et al.'s (2014) crustal study. Both of these upper mantle partial melt calculations estimate up to 1% partially molten. The NA13 and XL13P velocity anomalies are in line with those of Goes and van der Lee (2002) and Schutt and Humphreys (2004), reaffirming a 1% upper mantle partial melt beneath Yellowstone and the Snake River Plain. Based on the GG' profile in Figure 7, this strongly low-velocity anomaly in the uppermost mantle beneath Yellowstone continues at depth as a tall -1.5% *S* velocity anomaly that extends through the transition zone and into the lower mantle.

Nelson and Grand (2018) suggest that the Yellowstone hotspot has a plume origin, based on shear wave tomography. In the Nelson and Grand (2018) model, their hypothesized plume extends almost vertically, with a slightly southward dipping structure though the upper mantle and down to almost 1,000 km, where it then strongly dips to the southwest. Model NA13 might support this inference, but more strongly suggest a northward dipping structure though the upper mantle. Either way, the low velocities do not extend perfectly vertically through the transition, which shows high velocities beneath Yellowstone in the upper mantle and transition zone (Figure 7 transect GG' and HH'). However, the bottom half of the transition zone and the lower mantle down to 1,000 km indeed show similar low velocities as imaged by Nelson and Grand (2018). The non-vertical nature of the tall low-velocity anomaly connected to Yellowstone and the associated high velocities in the TZ beneath Yellowstone and the northern Basin and Range Province suggest that if Yellowstone has a plume origin, the plume may have been deflected by subducted lithosphere.

4.3. Southwestern United States

Figure 8 displays five profiles cross cutting the Rocky Mountain front that separates the tectonically active western United States from the relatively stable central and eastern United States. On close inspection, the velocity boundaries in the mantle, do not coincide directly along the Rocky Mountain front, and instead deviate to the west along the JJ' and KK' profiles, and to the east along LL', MM', and NN'. The intricate velocity pattern can be observed on a map at 150 km depth (Figure 8). At this depth, three high-velocity features extend from the stable Great Plains to the western US through the Columbia Plateau beneath Idaho, through the Wyoming craton and Colorado Plateau, and through northwestern and central Texas. The northern two high velocity features are separated along the Snake River Plain, in association with the reworking of the mantle in relation to the Yellowstone hotspot.

The southern two high V_s features are separated by lower velocities beneath the eastern margin of the Colorado Plateau (Figure 8), where the San Juan Mountains are. Similar P velocity anomalies were observed by Karlstrom et al. (2012). This separation in the uppermost mantle of high velocities in the Colorado Plateau region appears to align along the SW-NE trending border between the Northern and Southern Yavapai provinces, that can further be defined to the east by the Colorado mineral belt, which trends SW-NE just north of the San Juan volcanic region as noted in Figure 8. A joint inversion for 1D Vs profiles by Bailey et al. (2012) also shows the faster anomaly beneath the Colorado Plateau. They suggest that this thick, cool lithospheric root is at least partially responsible for the buoyancy and topography of the Colorado Plateau. NA13 models a faster anomaly to the south, than the Bailey et al. (2012) model. In addition, the NA13 anomalies extend slightly deeper, to ~200 km, as opposed to ~150 km in the Bailey et al. (2012) model.

Beneath southern Colorado, *S* velocities are less high, in alignment with higher expected temperatures, as modeled by Reiter (2008). Higher temperatures in this region would result in lower *S* velocities. The lowest *S*-velocities in this region are beneath the Southern Rocky Mountains and the San Juan volcanic, as seen in Figure 8, profile MM'. Indentations into the high velocity anomalies beneath the Colorado Plateau on the eastern side align with the Aspen Anomaly in the north, south to the San Juan anomaly, then to the Jemez lineament to the southeast, as seen at 150 km depth in Figure 8. The velocities in NA13 agree with those



modeled in a 2D transect by Gao et al. (2004), with higher velocities in the uppermost mantle beneath northwestern New Mexico, and lower velocities in central and southern New Mexico related to the Rio Grande Rift. Near the southern New Mexico and Texas border, high velocities are imaged again (transect N-N'), which Gao et al. (2004) associate with a downwelling in the mantle, associated with convection processes. As seen in profile NN', the mantle velocity boundary between the tectonically active western US and the North American craton appears to align with the surficial boundary.

4.4. Central and Eastern United States

Anomalies in mantle structure are also observed beneath the eastern United States, which is less well resolved in NA13 as mantle structure beneath the western USA. Resolution in the eastern United States is not as high as in the central and western US, due to data density, but features ~300 km diameter can still be resolved (Figure 5).

Maps of model NA13 in Figure 9 reveal heterogeneity in the mantle beneath the tectonically stable central and eastern US. Along the OO' profile through the Great Plains close to the Rocky Mountains, the *S* velocity undulates with a slow-fast-slow-fast pattern from north to south, indicating complex variation at the western edge of the North American craton. The PP' profile exhibits an east dipping low-velocity anomaly from New Mexico and northern Texas into the Great Plains to western Oklahoma in the Central Lowland. The low velocity anomalies just above the transition zone could most likely be related to mineralogical remnants of the Oklahoma Aulacogen or the thickening of the sub-lithospheric asthenosphere in conjunction with the thickening lithosphere. South of the Oklahoma Aulacogen which possibly formed during rifting of the proto-Atlantic (Burke & Dewey, 1973), the majority of Texas displays fast anomalies to 200 km depth as seen in the map view in Figure 9, representative of a more stable lithosphere, but is bordered to the west by lower velocities coincident in region to the Delaware Aulacogen.

Low velocities extend beneath Arkansas (profile PP') from the bottom of the 200-km thick lithosphere and the 410-km discontinuity, and beneath the New Madrid Seismic Zone at the northernmost point of the Coastal Plain up until the Interior Low Plateaus (Figure 1). The three-dimensional structure is further explored in QQ', which cuts through the NMSZ to the NNE. Here, the eastern Texas and Louisiana portion of the Coastal Plains display low velocities throughout much of the upper mantle. A northerly elongated slow anomaly hoovers above the transition zone and underlies the Mississippi Embayment and the New Madrid Seismic Zone, up until the Central Lowlands.

The mantle beneath the eastern Coastal Plains is characterized by low velocity anomalies throughout the upper mantle and transition zone (profile TT'). Strong low velocity anomalies are additionally observed beneath New England (profiles RR' and TT') and could be attributed to the thermal Northern Appalachian Anomaly (Menke et al., 2016), although alternate studies link a similarly placed low velocity anomaly to remnants of the Great Meteor hotspot track (Taylor & Fitzgerald, 2010). Two profiles along the MOMA (Wysession et al., 1996) (RR') and FLED (Wysession & Fischer, 2001) station arrays (SS') demonstrate velocity contrasts across the Appalachian Mountains between Precambrian lithosphere on the left of the profiles and that beneath the Paleozoic margins in the right of the profiles. The contrast is more pronounced along MOMA in the north than FLED in the south, but both are weaker, with contrasts of $2\% dV_s/100$ km and $\sim 1\% dV_s/100$ km than the ones across the Rocky Mountains. Profile TT' crosses the path of the Virginia upper mantle anomaly, which can be seen around 300 km beneath the West Virginia and Virginia border. This anomaly has been suggested to be related to the ~ 48 Ma Highland County, Virginia Volcanics (Mazza et al., 2014; Schmandt & Lin, 2014), possibly related to a lithospheric delamination.

Low velocity anomalies are also imaged in the transition zone and lower mantle beneath the east coast. These features were proposed by Van der Lee et al. (2008) to represent a wet upwelling, as the velocity anomalies cannot be as easily explained through variations in temperature or composition other than water. These low velocity anomalies extend down to the top of the lower mantle and possibly originate from dehydration of dense hydrous magnesium silicates in the subducting Farallon Plate.

5. Conclusions

The joint inversion of teleseismic *S* wave delay times, regional waveform fits, and constraints on Moho depth, resulted in an updated North American tomographic model, NA13. This model combines several decades of seismic data used for early North American seismic models (NA95, NA04, and NA07) and includes a substantial portion of USArray data. The joint model combines the merits of body waves' substantive lateral resolution and regional wave trains' substantive vertical resolution, and is superior to models based on one of these data sets. Model NA13 better reproduces the observed station-average delay time patterns between the coasts than a variety of tomographic models (Lou & van der Lee, 2014). It also sharpened velocity contrasts between the western, central, and eastern US.

In the western US, the subducting Juan de Fuca and Gorda slabs are clearly imaged, consistent with previous body-wave tomographic models (e.g., Burdick et al., 2012, 2014; James et al., 2011; Obrebski et al., 2010; Schmandt & Humphreys, 2010; Sigloch, 2011). Beneath the Snake River Plain and Yellowstone, very strong low velocity anomalies are imaged. In NA13, the peak anomaly beneath the Yellowstone Caldera is increased to -14%, enhancing the strong low velocity anomaly imaged by Schmandt and Humphreys (2010, 2011) and Tian et al. (2011). This large anomaly is too large to be solely attributed to thermal anomalies (Cammarano et al., 2003; Goes & van der Lee, 2002) implying that partial melting is likely present beneath Yellowstone. These strong low velocity anomalies at shallow depths beneath Yellowstone are associated with weaker, deeper low-velocity anomalies, suggesting a continuous plume conduit down to the lower mantle. This conduit is not vertical and appears to be deflected by higher velocity structures related to slab segments or continental lithosphere. The contrast between low velocities in the uppermost mantle beneath Yellowstone and those of the bordering Wyoming craton are over 17%.

Between the western and central US, the Rocky Mountain Front is defined by sharp velocity contrasts. The neutral anomaly contour within the mantle does not coincide with surface geological boundaries, deviating westwards in the north and eastwards at central latitudes. In the southernmost portion of the USA, mantle and crustal boundaries appear more aligned.

The complexity of the interface between the tectonically active and stable portions of the continent suggest that different processes and interactions are occurring between the surface and mantle, and cannot always be directly linked geographically. The relationship between surficial and upper mantle deformation is complex and highly dependent on regional and local variations in lithosphere strength. Seismic velocity anomalies reveal that the interaction of the North American craton and the tectonically active western United States can be influenced by local features such as Yellowstone and the Aspen Anomaly, as well as deeper features such as the topography of the 410 km discontinuity.

In the central and eastern United States, mantle is also heterogeneous despite being tectonically stable compared to the western United States. For example, deep low velocity anomalies were identified from beneath eastern Texas and Louisiana portion of the Coastal Plain to the Mississippi Embayment and the New Madrid Seismic Zone. The New England anomaly (Tian et al., 2011; Van der Lee & Frederiksen, 2005) and the eastern US anomaly (Van der Lee et al., 2008) are imaged with low velocities within the upper mantle and the transition zone. Overall, NA13 improves upon previous tomographic models by using both surface and body waves over a wide range of distances to constrain mantle structure. This improves the amplitude and location placement of velocity anomalies, and extends tomographic regional models deeper, with reliable results below the transition zone and into the lower mantle.

Data Availability Statement

Model NA13 is publically assessable through the IRIS data repository (https//doi.org/10.17611/dp/emc.2021.na13.1).



Acknowledgments

Seismic data was accessed through the Incorporated Research Institutions for Seismology (IRIS). Thank you to the National Science Foundation for funding the EarthScope and USArray initiatives for data collection and dissemination. Funding for this research was partially granted by the National Science Foundation grants EAR-0346200, and EAR-0645752.

References

- Atwater, T. (1970). Implications of plate tectonics for the Cenozoic tectonic evolution of western North America. *Geological Society of America Bulletin*, *81*, 3513–3536. https://doi.org/10.1130/0016-7606(1970)81[3513:ioptft]2.0.co;2
- Atwater, T. (1989). Plate tectonic history of the northeast Pacific and western North America. In E. Winterer, D. Hussong, & R. Decker (Eds.), *The eastern Pacific Ocean and Hawaii: The Geology of North America* (pp. 21–72). Geological Society of America.
- Bailey, I. W., Miller, M. S., Liu, K., & Levander, A. (2012). V_s and density structure beneath the Colorado Plateau constrained by gravity anomalies and joint inversions of receiver function and phase velocity data. *Journal of Geophysical Research*, *117*, B02313. https://doi. org/10.1029/2011JB008522
- Bally, A., Scotese, C., & Ross, M. (1989). North America; Plate-tectonic setting and tectonic elements. In A. Bally & A. Palmer (Eds.), *The Geology of North America An overview: The Geology of North America* (pp. 1–16). Geological Society of America.
- Bassin, C., Laske, G., & Masters, G. (2000). The current limits of resolution for surface wave tomography in North America. Eos, Transactions American Geophysical Union, 81, F897.
- Baumgardner, J. R., & Frederickson, P. O. (1985). Icosahedral discretization of the two-sphere. SIAM Journal on Numerical Analysis, 22, 1107–1115. https://doi.org/10.1137/0722066
- Bedle, H. (2008). Studies on the S-velocity structure of the North American upper mantle (Ph.D. thesis). Northwestern University.

Bedle, H., & van der Lee, S. (2009). S velocity variations beneath North America. Journal of Geophysical Research, 114, B07308. https://doi. org/10.1029/2008JB005949

Bensen, G. D., Ritzwoller, M. H., & Yang, Y. (2009). A 3D shear velocity model of the crust and uppermost mantle beneath the United States from ambient seismic noise. *Geophysical Journal International*, 177(3), 1177–1196. https://doi.org/10.1111/j.1365-246x.2009.04125.x
Bleeker, W. (2003). The late Archean record: A puzzle in ca. 35 pieces. *Lithos*, 71, 99–134. https://doi.org/10.1016/j.lithos.2003.07.003

- Bollmann, T. A., van der Lee, S., Frederiksen, A. W., Wolin, E., Revenaugh, J., Wiens, D. A., et al. (2019). P-wave teleseismic traveltime tomography of the North American Midcontinent. *Journal of Geophysical Research: Solid Earth*, 124, 1725–1742. https://doi. org/10.1029/2018JB016627
- Boschi, L., & Ekstrom, R. (2002). New images of the Earth's upper mantle from measurements of surface wave phase velocity anomalies. Journal of Geophysical Research, 107, ESE 1-1-ESE 1-14. https://doi.org/10.1029/2000JB000059
- Burdick, S., Li, C., Martynov, V., Cox, T., Eakins, J., Mulder, T., et al. (2008). Upper mantle heterogeneity beneath North America from travel time tomography with global and USArray transportable array data. *Seismological Research Letters*, 79, 384–392. https://doi. org/10.1785/gssrl.79.3.384
- Burdick, S., van der Hilst, R. D., Vernon, F. L., Martynov, V., Cox, T., Eakins, J., et al. (2012). Model update March 2011: Upper mantle heterogeneity beneath North America from traveltime tomography with global and USArray Transportable Array data. *Seismological Research Letters*, 83, 23–28. https://doi.org/10.1785/gssrl.83.1.23
- Burdick, S., van der Hilst, R. D., Vernon, F. L., Martynov, V., Cox, T., Eakins, J., et al. (2014). Model update January 2013: Upper mantle heterogeneity beneath North America from travel-time tomography with global and USArray Transportable Array data. *Seismological Research Letters*, 85(1), 77–81. https://doi.org/10.1785/0220130098
- Burke, K., & Dewey, J. (1973). Plume-generated triple junctions: Key indicators in applying plate tectonics to old rocks. The Journal of Geology, 81(4), 406–433. https://doi.org/10.1086/627882
- Cammarano, F., Goes, S., Vacher, P., & Giardini, D. (2003). Inferring upper-mantle temperatures from seismic velocities. *Physics of the Earth and Planetary Interiors*, 138, 197–222. https://doi.org/10.1016/S0031-9201(03)00156-0
- Cao, A., & Levander, A. (2010). High-resolution transition zone structures of the Gorda Slab beneath the western United States: Implication for deep water subduction. *Journal of Geophysical Research: Solid Earth*, 115(B7). https://doi.org/10.1029/2009jb006876
- Chang, S. J., van der Lee, S., Flanagan, M. P., Bedle, H., Marone, F., Matzel, E. M., et al. (2010). Joint inversion for three-dimensional S velocity mantle structure along the Tethyan margin. *Journal of Geophysical Research*, *115*. https://doi.org/10.1029/2009JB007204
- Chu, R., Helmberger, D., Sun, D., Jackson, J., & Zhu, L. (2010). Mushy magma beneath Yellowstone. *Geophysical Research Letters*, 37. https://doi.org/10.1029/2009GL041656
- Coney, P., & Reynolds, S. (1977). Cordilleran Beniof zones. Nature, 270, 403-406. https://doi.org/10.1038/270403a0
- Darbyshire, F. A., Eaton, D. W., Frederiksen, A. W., & Ertolahti, L. (2007). New insights into the lithosphere beneath the Superior Province from Rayleigh wave dispersion and receiver function analysis. *Geophysical Journal International*, *169*, 1043–1068. https://doi. org/10.1111/j.1365-246X.2006.03259.x
- Das, T., & Nolet, G. (1995). Crustal thickness estimation using high-frequency Rayleigh-waves. Geophysical Research Letters, 22, 539–542. https://doi.org/10.1029/94gl02845
- Davis, G. (1980). Problems of intraplate extensional tectonics, western United States. In B. C. Burchel, J. Oliver, & L. Silver (Eds.), Continental tectonics (pp. 84–95). National Academies Press.
- Delph, J. R., Levander, A., & Niu, F. (2018). Fluid controls on the heterogeneous seismic characteristics of the Cascadia margin. Geophysical Research Letters, 45, 11021–11029. https://doi.org/10.1029/2018GL079518
- Dueker, K., & Yuan, H. (2004). Upper mantle P-wave velocity structure from PASSCAL teleseismic transects across Idaho, Wyoming and Colorado. *Geophysical Research Letters*, 31. https://doi.org/10.1029/2004GL019476
- Durand, S., Debayle, E., Ricard, Y., Zaroli, C., & Lambotte, S. (2017). Confirmation of a change in the global shear velocity pattern at around 1,000 km depth. *Geophysical Journal International*, 211(3), 1628–1639. https://doi.org/10.1093/gji/ggx405
- Engebretson, D., Cox, A., & Gordon, R. (1984). Relative motions between oceanic plates of the Pacific basin. Journal of Geophysical Research, 89, 10291–10310. https://doi.org/10.1029/jb089ib12p10291
- Engebretson, D., Cox, A., & Gordon, R. (1985). Relative motions between oceanic and continental plates in the northern Pacific basin. *Geological Society of America Special Paper*, 206, 1–60. https://doi.org/10.1130/spe206-p1
- Feng, M., van der Lee, S., & Assumpcao, M. (2007). Upper mantle structure of South America from joint inversion of waveforms and fundamental mode group velocities of Rayleigh waves. *Journal of Geophysical Research*, 112. https://doi.org/10.1029/2006JB004449
- Foster, A., Darbyshire, F., & Schaeffer, A. (2020). Anisotropic structure of the central North American Craton surrounding the Mid-Continent Rift: Evidence from Rayleigh waves. *Precambrian Research*, 342, 342. https://doi.org/10.1016/j.precamres.2020.105662
- Frederiksen, A. W., Miong, S. K., Darbyshire, F. A., Eaton, D. W., Rondenay, S., & Sol, S. (2007). Lithospheric variations across the Superior Province, Ontario, Canada: Evidence from tomography and shear wave splitting. *Journal of Geophysical Research*, 112. https://doi.org/10.1029/2006JB004861
- Gao, W., Grand, S., Baldridge, W., Wilson, D., West, M., Ni, J., & Aster, R. (2004). Upper mantle convection beneath the central Rio Grande rift imaged by P and S wave tomography. *Journal of Geophysical Research*, *109*. https://doi.org/10.1029/2003JB002743

- Godey, S., Deschamps, F., Trampert, J., & Snieder, R. (2004). Thermal and compositional anomalies beneath the North American continent. *Journal of Geophysical Research*, 109. https://doi.org/10.1029/2002JB002263
- Goes, S., & van der Lee, S. (2002). Thermal structure of the North American uppermost mantle inferred from seismic tomography. Journal of Geophysical Research, 107. https://doi.org/10.1029/2000JB000049
- Golos, E. M., Fang, H., Yao, H., Zhang, H., Burdick, S., Vernon, F., et al. (2018). Shear wave tomography beneath the United States using a joint inversion of surface and body waves. *Journal of Geophysical Research: Solid Earth*, 123, 5169–5189. https://doi.org/10.1029/2017JB014894
- Grand, S. (1994). Mantle shear structure beneath the Americas and surrounding oceans. *Journal of Geophysical Research*, 99, 11591–11621. https://doi.org/10.1029/94JB00042
- Grand, S. (2002). Mantle shear-wave tomography and the fate of subducted slabs. *Philosophical Transactions of the Royal Society of London*, 360, 2475–2491. https://doi.org/10.1098/rsta.2002.1077
- Hoffman, P. (1988). United plates of America, the birth of a craton: Early Proterozoic assembly and growth of Laurentia. Annual Review of Earth and Planetary Sciences, 16, 543–603. https://doi.org/10.1146/annurev.ea.16.050188.002551
- Houser, C., Masters, G., Shearer, P., & Laske, G. (2008). Shear and compressional velocity models of the mantle from cluster analysis of long-period waveforms. *Geophysical Journal International*, 174, 195–212. https://doi.org/10.1111/j.1365-246X.2008.03763.x
- Huang, H.-H., Lin, F.-C., Schmandt, B., Farrell, J., Smith, R., & Tsai, V. C. (2014). The Yellowstone magmatic system from the mantle plume to the upper crust. *Science*, 348(6236), 773–776. https://doi.org/10.1126/science.aaa5648
- Humphreys, E., & Dueker, K. (1994). Physical state of the western US upper mantle. *Journal of Geophysical Research*, 99, 9635–9650. https://doi.org/10.1029/93JB02640
- Humphreys, E. D., Dueker, K. G., Schutt, D. L., & Smith, R. B. (2000). Beneath Yellowstone: Evaluating plume and nonplume models using teleseismic images of the upper mantle. *Geological Society of America Today*, 10, 1–7.
- James, D. E., Fouch, M. J., Carlson, R. W., & Roth, J. B. (2011). Slab fragmentation, edge flow and the origin of the Yellowstone hotspot track. *Earth and Planetary Science Letters*, 311, 124–135. https://doi.org/10.1016/j.epsl.2011.09.007
- Karlstrom, K., Coblentz, D., Dueker, K., Ouimet, W., Kirby, E., Van Wijk, J., et al. (2012). Mantle-driven dynamic uplift of the Rocky Mountains and Colorado Plateau and its surface response: Toward a unified hypothesis. *Lithosphere*, 4(1), 3–22. https://doi.org/10.1130/ L150.1
- Kennett, B. L. N., & Engdahl, E. R. (1991). Travel times for global earthquake location and phase association. Geophysical Journal International, 105, 429–465. https://doi.org/10.17611/DP/999180910.1111/j.1365-246x.1991.tb06724.x
- Laske, G., & Masters, G. (1998). Surface-wave polarization data and global anisotropic structure. *Geophysical Journal International*, *132*, 508–520. https://doi.org/10.1046/j.1365-246X.1998.00450.x
- Levander, A., Schmandt, B., Miller, M., Liu, K., Karlstrom, K. E., Crow, R. S., et al. (2011). Continuing Colorado plateau uplift by delamination-style convective lithospheric downwelling. *Nature*, 472, 461–465. https://doi.org/10.1038/nature10001
- Levander, A., Zelt, C., & Magnani, M. B. (2005). Crust and upper mantle velocity structure of the southern Rocky Mountains from the Jemez Lineament to the Cheyenne Belt. In K. E. Karlstrom & G. R. Keller (Eds.), *Rocky Mountain region: An evolving lithosphere: Tectonics, geochemistry, and geophysics* (Vol. 154, pp. 293–308). Geophysical Monograph Series. https://doi.org/10.1029/154GM22
- Levin, V., Long, M. D., Skryzalin, P., Li, Y., & López, I. (2018). Seismic evidence for a recently formed mantle upwelling beneath New England. Geology, 46, 87–90. https://doi.org/10.1130/g39641.1
- Li, B. (2003). Compressional and shear wave velocities of ringwoodite γ-Mg₂SiO₄ to 12 GPa. *American Mineralogist*, *88*, 1312–1317. https://doi.org/10.2138/am-2003-8-913
- Li, X., & Romanowicz, B. (1996). Global mantle shear velocity model developed using non-linear asymptotic coupling theory. Journal of Geophysical Research, 101, 22245–22272. https://doi.org/10.1029/96JB01306
- Lin, F. C., Moschetti, M. P., & Ritzwoller, M. H. (2008). Surface wave tomography of the western United States from ambient seismic noise: Rayleigh and Love wave phase velocity maps. *Geophysical Journal International*, 173, 281–298. https://doi. org/10.1111/j.1365-246X.2008.03720.x
- Lin, F. C., & Ritzwoller, M. H. (2011). Helmholtz surface wave tomography for isotropic and azimuthally anisotropic structure. *Geophysical Journal International*, 186, 1104–1120. https://doi.org/10.1111/j.1365-246X.2011.05070.x
- Liu, K., Levander, A., Niu, F., & Miller, M. S. (2011). Imaging crustal and upper mantle structure beneath the Colorado Plateau using finite frequency Rayleigh wave tomography. *Geochemistry, Geophysics, Geosystems, 12*. https://doi.org/10.1029/2011GC003611
- Long, M. D., Benoit, M. H., Aragon, J. C., & King, S. D. (2019). Seismic imaging of midcrustal structure beneath central and eastern North America: Possibly the elusive Grenville deformation? *Geology*, 47, 371–374. https://doi.org/10.1130/G46077.1
- Lou, X. (2013). Inferred and predicted seismic velocities of the North American mantle (Ph.D. thesis). Northwestern University.
- Lou, X., & van der Lee, S. (2014). Observed and predicted North American teleseismic delay times. *Earth and Planetary Sciences Letters*, 402, 6–15. https://doi.org/10.1016/j.epsl.2013.11.056
- Lou, X., van der Lee, S., & Lloyd, S. (2013). AIMBAT: A Python/Matplotlib tool for measuring teleseismic arrival times. Seismological Research Letters, 84, 85–93. https://doi.org/10.1785/0220120033
- Lowry, A. R., Becker, T. W., Buehler, J. S., Miller, M. S., Pérez-Gussinyé, M., Schutt, D. L., & Seunarine, L. L. (2013). Challenges (and promise) of in-situ lithospheric rheology from isostatic strength analyses. *EGU General Assembly* (Vol. 15). Abstract EGU2013-6248.
- Lowry, A. R., & Pérez-Gussinyé, M. (2011). The role of crustal quartz in controlling Cordilleran deformation. *Nature*, 471, 353–357. https://doi.org/10.1038/nature09912
- Lu, C., Grand, S., Lai, H., & Garnero, E. (2019). TX2019slab: A new P and S tomography model incorporating subducting slabs. Journal of Geophysical Research: Solid Earth, 124, 11549–11567. https://doi.org/10.1029/2019JB017448
- Marone, F., Gung, Y., & Romanowicz, B. (2007). Three-dimensional radial anisotropic structure of the North American upper mantle from inversion of surface waveform data. *Geophysical Journal International*, 171, 206–222. https://doi.org/10.1111/j.1365-246x.2007.03465.x
- Mazza, S. E., Gazel, E., Johnson, E. A., & Schmandt, B. (2014). Volcanoes of Virginia: A window into the post rift evolution of the eastern North American margin (Vol. 33). Geoprisms Newsletters.
- Menke, W., Skryzalin, P., Levin, V., Harper, T., Darbyshire, F., & Dong, T. (2016). The Northern Appalachian Anomaly: A modern asthenospheric upwelling. *Geophysical Research Letters*, 43, 10173–10179. https://doi.org/10.1002/2016GL070918
- Molnar, P., & Atwater, T. (1978). Interarc spreading and Cordilleran tectonics as alternates related to the age of subducted oceanic lithosphere. *Earth and Planetary Science Letters*, 41, 330–340. https://doi.org/10.1016/0012-821x(78)90187-5
- Nelson, P., & Grand, S. (2018). Lower-mantle plume beneath the Yellowstone hotspot revealed by core waves. *Nature Geoscience*, 11, 280–284. https://doi.org/10.1038/s41561-018-0075-y



- Nettles, M., & Dziewonski, A. M. (2008). Radially anisotropic shear velocity structure of the upper mantle globally and beneath North America. *Journal of Geophysical Research*, 113. https://doi.org/10.1029/2006JB004819
- Nolet, G. (1990). Partitioned waveform inversion and two-dimensional structure under the network of autonomously recording seismographs. *Journal of Geophysical Research*, 95, 8499–8512. https://doi.org/10.1029/JB095iB06p08499
- Obrebski, M., Allen, R. M., Pollitz, F., & Hung, S. H. (2011). Lithosphere-asthenosphere interaction beneath the western United States from the joint inversion of body-wave traveltimes and surface-wave phase velocities. *Geophysical Journal International*, *185*, 1003–1021. https://doi.org/10.1111/j.1365-246X.2011.04990.x
- Obrebski, M., Allen, R. M., Xue, M., & Hung, S. H. (2010). Slab-plume interaction beneath the Pacific Northwest. *Geophysical Research Letters*, 37. https://doi.org/10.1029/2010GL043489
- Paige, C., & Saunders, M. (1982a). LSQR: An algorithm for sparse linear equations and sparse least squares. ACM Transactions on Mathematical Software (TOMS), 8, 43–71. https://doi.org/10.1145/355984.355989
- Paige, C., & Saunders, M. (1982b). LSQR: Sparse linear equations and least squares problems. ACM Transactions on Mathematical Software (TOMS), 8, 195–209. https://doi.org/10.1145/355993.356000
- Panning, M. P., Lekic, V., & Romanowicz, B. A. (2010). Importance of crustal corrections in the development of a new global model of radial anisotropy. *Journal of Geophysical Research*, 115. https://doi.org/10.1029/2010JB007520
- Pollitz, F. F. (2008). Observations and interpretation of fundamental mode Rayleigh wavefields recorded by the Transportable Array (US-Array). Journal of Geophysical Research, 113. https://doi.org/10.1029/2007JB005556
- Pollitz, F. F., & Snoke, J. A. (2010). Rayleigh-wave phase-velocity maps and three-dimensional shear velocity structure of the western US from local non-plane surface wave tomography. *Geophysical Journal International*, 180, 1153–1169. https://doi. org/10.1111/j.1365-246X.2009.04441.x
- Reiter, M. (2008). Geothermal anomalies in the crust and upper mantle along Southern Rocky Mountain transitions. Geological Society of America Bulletin, 120(3–4), 431–441. https://doi.org/10.1130/b26198.1
- Ren, Y., Stutzmann, E., van der Hilst, R. D., & Besse, J. (2007). Understanding seismic heterogeneities in the lower mantle beneath the Americas from seismic tomography and plate tectonic history. *Journal of Geophysical Research*, 112. https://doi.org/10.1029/2005JB004154
- Ritsema, J., Deuss, A., van Heijst, H. J., & Woodhouse, J. H. (2011). S40RTS: A degree-40 shear-velocity model for the mantle from new Rayleigh wave dispersion, teleseismic traveltime and normal-mode splitting function measurements. *Geophysical Journal International*, 184, 1223–1236. https://doi.org/10.1111/j.1365-246X.2010.04884.x
- Ritsema, J., van Heijst, H., & Woodhouse, J. (2004). Global transition zone tomography. Journal of Geophysical Research, 109. https://doi. org/10.1029/2003JB002610
- Rogers, G., & Dragert, H. (2003). Episodic tremor and slip on the Cascadia subduction zone: The chatter of silent slip. *Science*, 300, 1942–1943. https://doi.org/10.1126/science.1084783
- Roth, J. B., Fouch, M. J., James, D. E., & Carlson, R. W. (2008). Three-dimensional seismic velocity structure of the northwestern United States. Geophysical Research Letters, 35. https://doi.org/10.1029/2008GL034669
- Schmandt, B., & Humphreys, E. (2010). Complex subduction and small-scale convection revealed by body-wave tomography of the western United States upper mantle. *Earth and Planetary Science Letters*, 297, 435–445. https://doi.org/10.1016/j.epsl.2010.06.047
- Schmandt, B., & Humphreys, E. (2011). Seismically imaged relict slab from the 55 Ma Siletzia accretion to the northwest United States. *Geology*, 39, 175–178. https://doi.org/10.1130/G31558.1
- Schmandt, B., & Lin, F. C. (2014). P and S wave tomography of the mantle beneath the United States. *Geophysical Research Letters*, 41, 6342–6349. https://doi.org/10.1002/2014GL061231
- Schmid, C., van der Lee, S., VanDecar, J. C., Engdahl, E. R., & Giardini, D. (2008). Three-dimensional S velocity of the mantle in the Africa-Eurasia plate boundary region from phase arrival times and regional waveforms. *Journal of Geophysical Research*, 113. https://doi. org/10.1029/2005JB004193
- Schutt, D., & Humphreys, E. (2004). P and S wave velocity and V_p/V_S in the wake of the Yellowstone hot spot. Journal of Geophysical Research: Solid Earth, 109. https://doi.org/10.1029/2003JB002442
- Schutt, D. L., Dueker, K., & Yuan, H. (2008). Crust and upper mantle velocity structure of the Yellowstone hot spot and surroundings. Journal of Geophysical Research, 113. https://doi.org/10.1029/2007JB005109
- Shearer, P., & Buehler, J. (2019). Imaging upper-mantle structure under USArray using long-period reflection seismology. Journal of Geophysical Letters: Solid Earth, 124(9), 9638–9652. https://doi.org/10.1029/2019JB017326
- Sigloch, K. (2011). Mantle provinces under North America from multifrequency P wave tomography. Geochemistry, Geophysics, Geosystems, 12. https://doi.org/10.1029/2010GC003421
- Sigloch, K., McQuarrie, N., & Nolet, G. (2008). Two-stage subduction history under North America inferred from multiple-frequency tomography. *Nature Geoscience*, 1, 458–462. https://doi.org/10.1038/ngeo231
- Simmons, N. A., Forte, A. M., Boschi, L., & Grand, S. P. (2010). GyPSuM: A joint tomographic model of mantle density and seismic wave speeds. Journal of Geophysical Research, 115. https://doi.org/10.1029/2010JB007631
- Simmons, N. A., Myers, S. C., Johannesson, G., & Matzel, E. (2012). LLNL-G3Dv3: Global P wave tomography model for improved regional and teleseismic travel time prediction. *Journal of Geophysical Research*, 117. https://doi.org/10.1029/2012JB009525
- Tanimoto, T., & Sheldrake, K. P. (2002). Three-dimensional S-wave velocity structure in southern California. Geophysical Research Letters, 29, 64-1–64-4. https://doi.org/10.1029/2001GL013486
- Tauzin, B., Bodin, T., Debayle, E., Perrillat, J. P., & Reynard, B. (2016). Multi-mode conversion imaging of the subducted Gorda and Juan de Fuca plates below the North American continent. *Earth and Planetary Science Letters*, 440, 135–146. https://doi.org/10.1016/j. epsl.2016.01.036
- Tauzin, B., van der Hilst, R. D., Wittlinger, G., & Ricard, Y. (2013). Multiple transition zone seismic discontinuities and low velocity layers below western United States. *Journal of Geophysical Research: Solid Earth*, 118(5), 2307–2322. https://doi.org/10.1002/jgrb.50182
- Taylor, J. A., & Fitzgerald, P. G. (2010). Post-Jurassic thermal history and exhumation of the Eastern Adirondack Mountains associated with movement over the Great Meteor Hotspot: Constraints from low-temperature thermochronology. *Geological Society of America Bulletin*, 123, 412–426. https://doi.org/10.1130/B30138.1

Thompson, A. B. (1992). Water in the Earth's upper mantle. Nature, 358, 295-302. https://doi.org/10.1038/358295a0

- Tian, Y., Sigloch, K., & Nolet, G. (2009). Multiple-frequency SH-wave tomography of the western US upper mantle. *Geophysical Journal International*, 178, 1384–1402. https://doi.org/10.1111/j.1365-246X.2009.04225.x
- Tian, Y., & Zhao, D. (2012). P-wave tomography of the western United States: Insight into the Yellowstone hotspot and the Juan de Fuca slab. *Physics of the Earth and Planetary Interiors*, 200, 72–84. https://doi.org/10.1016/j.pepi.2012.04.004



- Tian, Y., Zhou, Y., Sigloch, K., Nolet, G., & Laske, G. (2011). Structure of North American mantle constrained by simultaneous inversion of multiple-frequency SH, SS, and Love waves. *Journal of Geophysical Research*, *116*. https://doi.org/10.1029/2010JB007704
- Van der Hilst, R., Widiyantoro, S., & Engdahl, E. (1997). Evidence for deep mantle circulation from global tomography. *Nature*, 386, 578–584. https://doi.org/10.1038/386578a0
- Van der Lee, S., & Frederiksen, A. (2005). Surface wave tomography applied to the North American upper mantle. In A. Levander & G. Nolet (Eds.), Seismic Earth: Array analysis of broadband seismograms. Geophysical Monograph Series (Vol. 157, pp. 67–80). https://doi.org/10.1029/157GM05
- Van der Lee, S., & Nolet, G. (1997a). Seismic image of the subducted trailing fragments of the Farallon plate. *Nature*, 386, 266–269. https://doi.org/10.1038/386266a0
- Van der Lee, S., & Nolet, G. (1997b). Upper mantle S velocity structure of North America. Journal of Geophysical Research, 102, 22815– 22838. https://doi.org/10.1029/97JB01168
- Van der Lee, S., Regenauer-Lieb, K., & Yuen, D. A. (2008). The role of water in connecting past and future episodes of subduction. Earth and Planetary Science Letters, 273, 15–27. https://doi.org/10.1016/j.epsl.2008.04.041
- Villemaire, M., Darbyshire, F. A., & Bastow, I. D. (2012). P-wave tomography of eastern North America: Evidence for mantle evolution from Archean to Phanerozoic, and modification during subsequent hot spot tectonism. *Journal of Geophysical Research*, 117. https://doi.org/10.1029/2012JB009639
- Wagner, L., Forsyth, D. W., Fouch, M. J., & James, D. E. (2010). Detailed three-dimensional shear wave velocity structure of the northwestern United States from Rayleigh wave tomography. *Earth and Planetary Science Letters*, 299, 273–284. https://doi.org/10.1016/j. epsl.2010.09.005
- Wang, Z., & Dahlen, F. (1995). Spherical-spline parameterization of three-dimensional Earth models. *Geophysical Research Letters*, 22, 3099–3102. https://doi.org/10.1029/95GL03080
- West, M., Ni, J., Baldridge, W., Wilson, D., Aster, R., Gao, W., & Grand, S. (2004). Crust and upper mantle shear wave structure of the southwest United States: Implications for rifting and support for high elevation. *Journal of Geophysical Research*, 109. https://doi. org/10.1029/2003JB002575
- Wysession, M. E., & Fischer, K. M. (2001). The Florida to Edmonton broadband experiment. International Federation of Digital Seismograph Networks, Dataset/Seismic Network. https://doi.org/10.7914/SN/XR_2001
- Wysession, M. E., Fischer, K. M., Clarke, T. J., Al-eqabi, G. I., Fouch, M. J., Shore, P. J., et al. (1996). Slicing into the Earth. Eos Transactions American Geophysical Union, 77(49), 477–488.
- Xue, M., & Allen, R. A. (2007). The fate of the Juan de Fuca plate: Implications for a Yellowstone plume head. *Earth and Planetary Science Letters*, 264, 266–276. https://doi.org/10.1016/j.epsl.2007.09.047
- Xue, M., & Allen, R. M. (2010). Mantle structure beneath the western United States and its implications for convection processes. Journal of Geophysical Research, 115. https://doi.org/10.1029/2008JB006079
- Yang, X., & Gao, H. (2018). Full-wave seismic tomography in the northeastern United States: New insights into the uplift mechanism of the Adirondack Mountains. *Geophysical Research Letters*, 45, 5992–6000. https://doi.org/10.1029/2018GL078438
- Yang, Y., Ritzwoller, M. H., Lin, F. C., Moschetti, M. P., & Shapiro, N. M. (2008). Structure of the crust and uppermost mantle beneath the western United States revealed by ambient noise and earthquake tomography. *Journal of Geophysical Research*, 113. https://doi. org/10.1029/2008JB005833
- Yuan, H., & Dueker, K. (2005a). Teleseismic P-wave tomogram of the Yellowstone plume. *Geophysical Research Letters*, 32. https://doi. org/10.1029/2004GL022056
- Yuan, H., & Dueker, K. (2005b). Upper mantle tomographic V_p and V_s images of the Rocky Mountains in Wyoming, Colorado and New Mexico: Evidence for a thick heterogeneous chemical lithosphere. In K. E. Karlstrom & G. R. Keller (Eds.), Rocky Mountain region: An evolving lithosphere: Tectonics, geochemistry, and geophysics. Geophysical Monograph Series (Vol. 154, pp. 329–345). https://doi.org/10.1029/154GM25
- Zandt, G., Gilbert, H., Owens, T. J., Ducea, M., Saleeby, J., & Jones, C. H. (2004). Active foundering of a continental arc root beneath the southern Sierra Nevada in California. *Nature*, 431, 41–46. https://doi.org/10.1038/nature02847
- Zhao, D. (2004). Global tomographic images of mantle plumes and subducting slabs: Insight into deep Earth dynamics. *Physics of the Earth and Planetary Interiors*, 146, 3–34. https://doi.org/10.1016/j.pepi.2003.07.032
- Zhao, K., Luo, Y., & Xie, J. (2017). Broad-band Rayleigh wave phase velocity maps (10–150 s) across the United States from ambient noise data. Geophysical Journal International, 108(2). https://doi.org/10.1093/gji/ggw460